\documentclass[leqno,12 pt]{amsart}
\usepackage{amsfonts,amsmath,amssymb,graphicx,psfrag}
\usepackage{mathrsfs,amsmath}
\addtolength{\textwidth}{3cm}
\addtolength{\textheight}{2cm}
\addtolength{\oddsidemargin}{-1cm}
\addtolength{\evensidemargin}{-1cm}

\newtheorem{theorem}[equation]{Theorem}
\newtheorem{corollary}[equation]{Corollary}
\newtheorem{lemma}[equation]{Lemma}
\newtheorem{proposition}[equation]{Proposition}
\newtheorem{remark}[equation]{Remark}
\newtheorem{definition}{Definition}[section]
\newtheorem{example}{Example}[section]
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{99}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{An} G. Andrews, R. Askey and Q. Ranjan,{\it  special function}, Cambridge University Press, (1999).

%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Berg}  C. Berg, Stieltjes-Pick-Bernstein-Schoenberg and their connection to comeplete
monotonicity. Pages 15–-- in Positive Definite Functions: From Schoenberg to Space-Time challenges. J. Mateu and E. Porcu eds. Castellon de la Plana 2008.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Bloom}Bloom W. R, Heyer H. Harmonic analysis of probability measures on hypergroups, In De Gruyter
Studies in Mathematics, 20, (H. Bauer, J.L. Kazdan and E. Zehnder, Editors), De Gruyter, Berlin, New York; 1994.
\bibitem{Bou}Bouzeffour F, Garayev M. On the fractional Bessel operator. Integral Transform Spec Funct. 2022;33(3):230--246.
%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Cholowinski}Cholewinski, Frank M. et al. A necessary and sufficient condition for the representation of a function as a Hankel-Stieltjes transform. Studia Mathematica 36 (1970): 269--274.
%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Chebli}Ch\'{e}bli H. Op\'{e}rateurs de translation g\'{e}n\'{e}ralis\'{e}e et semi-groupe de
convolution, Lectures notes. 1974;404:35--59.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Dimovski2} Dimovski IH, Kiryakova VS.  Transmutations, convolutions and
fractional powers of Bessel-type operators via Meijer’s G-function, in:
“Complex Analysis and Applications. 1983;(83):45--66.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Main1}F. Mainardi, Y. Luchko, G. Pagnini, The fundamental solution of the space-time fractional diffusion equation, Fractional Calculus and Applied Analysis
4 (2) (2001) 153--192.
\bibitem{Main3}Mainardi F. Fractional calculus and waves in linear viscoelasticity: An introduction to
mathematical models. World Scientific; 2010.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Lyakhov}  Lyakhov, L.N., “A class of hypersingular integrals”, Dokl. Math.1991;42(3):765--769.
\bibitem{Lyakhov3} Lyakhov LN. “Inversion of Riesz B-potentials”, Dokl. Math.1992; 44(3): 717--720

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{Rudin} W. Rudin, Real and Complex Analysis, 2nd ed. New York: McGraw-Hill, 1974.
\bibitem{Sc} I.J. Schoenberg, Metric spaces and completely monotonic functions. Ann. Math. 39, (1938), 811-841.
\bibitem{Trigueb} R. M. Trigub, Absolute convergence of Fourier integrals, summability of Fourier series, and approximation of functions by polynomials on a torus, Izv. Akad. Nauk SSSR, Ser. Mat., 44, No. 6, 1378--1408 (1980).

\bibitem{Trimeche}Trim\'{e}che K.  Generalized Wavelets and Hypergroups. Gordon \& Breach, New York; 1997.






\bibitem{Kiryakova} Kiryakova V. Generalized Fractional Calculus and Applications.
Pitman Res. Notes Math. 301, Longman Scientific \& Technical, Harlow,
Co-publ. John Wiley, New York; 1994.

\bibitem{Kilbas} Kilbas A, Srivastava H, Trujillo J. Theory and applications of fractional differential equations, 2006.

\bibitem{8} G.N. Watson, {\it A treatise on the theory of Bessel functions}, Cambridge University Press (1990).

\bibitem{We} H. Wendland , {\it Scattered data approximations },
 Cambridge University Press. , Cambridge, 2005.

 \end{thebibliography}

\newcommand\nonu{\nonumber}
\newcommand\dstyle\displaystyle
\newcommand\sa{\smallskipamount}
\newcommand\ma{\medskipamount}
\newcommand\ba{\bigskipamount}
\newcommand\sLP{\\[\sa]}
\newcommand\mLP{\\[\ma]}
\newcommand\bLP{\\[\ba]}
\newcommand\bPP{\\[\ba]\indent}
\newcommand\RR{\mathbb{R}}
\newcommand\ZZ{\mathbb{Z}}
\newcommand\FSE{{\cal E}}
\newcommand\FSJ{{\cal J}}
\newcommand\goH{\mathfrak{H}}
\newcommand\al\alpha
\newcommand\be\beta
\newcommand\Ga{\Gamma}
\newcommand\ka\kappa
\newcommand\1{\text{\rm 1}\hspace{-.95mm}\text{\rm I}}
\newcommand\C{\mathbb{C}}
\renewcommand\epsilon{\varepsilon}
\newcommand\N{\mathbb{N}}
\newcommand\R{\mathbb{R}}
\newcommand\ssf{\hspace{.25mm}}
\newcommand\ssb{\hspace{-.25mm}}
\newcommand\supp{\operatorname{supp}}
\newcommand\Z{\mathbb{Z}}
\newcommand\const{\operatorname{const.}}
\newcommand\iy\infty
\newcommand\lan{\langle}
\newcommand\ran{\rangle}
\newcommand\wh{\widehat}
\newcommand\wt{\widetilde}
\newcommand\LHS{left-hand side}
\newcommand\RHS{right-hand side}
\newcommand\Equiva{\Longleftrightarrow}
\newcommand{\hyp}[5]{\,\mbox{}_{#1}F_{#2}\!\left(
  \genfrac{}{}{0pt}{}{#3}{#4};#5\right)}
\newcommand{\qhyp}[5]{\,\mbox{}_{#1}\phi_{#2}\!\left(
  \genfrac{}{}{0pt}{}{#3}{#4};#5\right)}
\newcommand\bma{\begin{pmatrix}}
\newcommand\ema{\end{pmatrix}}
\newcommand{\sign}{\text{sgn}}


\title{ Continuation of Radial Positive Definite Functions and Their Characterization}

\author[Fethi Bouzeffour]{ Fethi  Bouzeffour }
\address{Department of mathematics, College of Sciences,
King Saud University, P.O. Box 2455, Riyadh 11451, Saudi Arabia.}
\email{fbouzaffour@ksu.edu.sa}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\subjclass[2010]{Primary\,: 33C10,46F12
.
Secondary\,: 33C67, 26A33}

\keywords{Positive definite functions, Completely monotone functions, Fractional integral and derivative, Fractional diffusion equation}
\date{\today}

\begin{abstract}This paper investigates the continuation of radial positive definite functions and their characterization beyond integer dimensions. We present a comprehensive characterization of radial positive definite functions using the Riemann-Liouville fractional integral and fractional Caputo derivatives. Furthermore, a secondary characterization based on the Bernstein characterization of completely monotone functions is introduced. The practical relevance of this study is demonstrated by investigating the positivity of the fundamental solution of the space-fractional Bessel diffusion equation.
\end{abstract}

\maketitle
\section{Introduction}
Radial positive definite functions play a crucial role in numerous mathematical fields, finding applications in functional analysis, probability theory, signal processing, and more \cite{Trigueb, We, Main3}. Although traditionally studied in Euclidean spaces, there has been growing interest in extending these functions beyond integer dimensions. Notably, the works of Cholewinski et al. \cite{Cholowinski} and Chebli \cite{Chebli} have explored the continuation of radial positive definite functions associated with Bessel operators of arbitrary order and a family of singular regular differential operators, including the Bessel and Jacobi operators, see also \cite{Trimeche}.

Fractional calculus provides a powerful framework for analyzing functions with non-local and long-range dependencies, making it suitable for studying the continuation of positive definite functions beyond integer dimensions \cite{Kilbas,Trigueb, Main1, Kiryakova, Main3,Dimovski2, Lyakhov,Lyakhov3}. By examining the interplay between fractional derivatives and the non-integer continuation of radial positive functions, valuable insights can be gained into their nature and structure.

This study aims to delve into the continuation of radial positive definite functions and their characterization using fractional derivatives, building upon the foundational works of Cholewinski et al. and Chebli. Furthermore, an alternative characterization based on completely monotone functions is presented. Completely monotone functions possess significant properties and are extensively studied in various mathematical areas \cite{Sc,Berg}.\\
To demonstrate the practical relevance of the findings, this research explores the application of these extended positive definite functions in the context of the space fractional diffusion equation. By utilizing the obtained characterization, the positivity of the fundamental solution is investigated, highlighting the practical implications and potential applications of the continued radial positive definite functions.\\
The paper is structured as follows. In Section 2, an overview and introduction to the notations and key facts related to the Fourier-Bessel transform and the Delsarte translation are provided.

Section 3 delves into the concept of radial positive definite functions, with a particular focus on the renowned Bochner and Schoenberg theorems. The continuation of radial definite positive functions is also explored in this section.

In Section 4, a characterization of the representation of functions on the interval $(0, \infty)$ as Fourier-Bessel transforms using Riemann-Liouville integral and Caputo fractional derivative is presented.

Section 5 focuses on the characterization of the class $\mathcal{P}_\ka$ by leveraging Bernstein's theorem for completely monotone functions.

In the last section, the positivity of the fundamental solution of the space-fractional Bessel diffusion equation is demonstrated






\section{Preliminaries}
This section serves as an introduction to the notations and key facts related to the Fourier-Bessel transform and the Delsarte translation. These concepts are essential for understanding the subsequent analysis and characterization of radial positive definite functions beyond integer dimensions.

First, we introduce the normalized Bessel function, denoted as $\mathscr{J}_\ka(x)$, which plays a fundamental role in the sequel. The normalized Bessel function defined by
 \[
\mathscr{J}_\ka(x):=\Ga(\ka+1)\,(2/x)^\ka\,J_\ka(x), \quad \ka>-1,
\]
where  $J_\ka(\cdot)$ is the Bessel function, see \cite[(10.16.9)]{8}. Then
\[
\mathscr{J}_\ka(x)=
\sum_{k=0}^\iy\frac{(-\tfrac14 x^2)^k}{(\ka+1)_k\,k!}
=\hyp01-{\ka+1}{-\tfrac14x^2}\qquad(\kappa>-1).
\]
The normalized Bessel function emerges as the unique solution of the eigenvalue problem associated with the Bessel equation. Specifically, the functions $x\rightarrow \mathscr{J}_\ka(\lambda x)$ is the unique solution of the eigenvalue problem
\cite[(10.13.5)]{8}
\begin{equation*}
\begin{cases}
&\mathscr{B}_\ka u(x) =-\lambda^2u(x),\mLP
&\dstyle
u(0)=1,\quad u'(0)=0.
\end{cases}
\label{1}
\end{equation*}
where $\mathscr{B}_\ka$ is the Bessel operator
\begin{equation*}
\mathscr{B}_\ka:=\frac{d^2}{d
x^2}+\frac{2\ka+1}x\,\frac {d}{dx},\quad \ka \geq -1/2.
\end{equation*}
The function $\mathscr{J}_\kappa(\cdot)$ is an even entire analytic function and we have the simple
 special cases
\[
\mathscr{J}_{-1/2}(x)=\cos x,\quad
\mathscr{J}_{1/2}(x)=\frac{\sin x}x\,.
\]
By \cite[\S 7.21]{8}), we have the asymptotic expansion of the normalized Bessel function
\begin{align}\label{asm}
  \mathscr{J}_\ka(x)&=\cos(x-\frac{\ka\pi}{2}-\frac{\pi}{4})+O(\frac{1}{x}),\quad x\rightarrow \infty,\quad  \ka > -\frac{1}{2}.
\end{align}
We denote by  $L^p_\ka(0,\,\iy)$ $(1\leq p),$ the Lebesgue space associated with the measure
\begin{equation*}\sigma_\ka(dx)=\frac{x^{2\ka+1}}{2^{\ka+1}\Gamma(\ka+1)}\,dx\end{equation*}  and  by $\|f\|_{\ka,p}$ the
usual norm given by
\begin{align*}
&\|f\|_{\ka,p}=\Big(\int_{0}^\iy|f(x)|^p\,\sigma_\ka(dx)\Big)^{1/p}.
\end{align*}
The Fourier-Bessel transform $\mathscr{F}_\ka f$ of $f\in L^1_{\ka}(0,\,\infty)$ is defined as:
\begin{equation}\label{fb1}
\mathscr{F}_\ka f(x) = \int_{0}^\infty f
(t)  \mathscr{J}_\ka(tx)  \sigma_\ka(dx), \quad \ka\geq -1/2.
\end{equation}
This integral transform  plays a similar role as the Fourier transform in the Euclidean one. In particular,  it can be extended to an isometry of $L^2_\ka(0,\,\infty)$
and for any $f\in L^1_{\ka}(0,\,\infty)\cap L^2_{\ka}(0,\,\infty)$ we have
\begin{equation}
\int_{0}^\iy  |f(x)|^2\,
\sigma_\ka(dx)=\int_{0}^\iy |\mathscr{F}_\ka f(t)|^2\,
\,\sigma_\ka(dt)
\end{equation}
and its inverse is given by
\begin{equation}\label{inv1}
f(x)=\int_{0}^\iy \mathscr{F}_\ka f(t)\,
\mathcal{J}_\ka(tx)\,\sigma_\ka(dt).
\end{equation}
Next, we discuss the generalized translation operator associated with the Bessel operator. This operator is denoted as $\tau_\ka^x,$  and acts on functions  $f\in L^1_\ka(0,\,\infty)$
as follows \cite[\S 3.4.1]{Bloom}
\begin{equation}\label{pB}
\tau^x_\ka f(y)=\left\{
  \begin{array}{l l} \int_{0}^{\pi}f(\sqrt{x^2+y^2+ 2xy\cos\theta} )\sin^{2\ka} \theta \, d\theta,\quad \text{if}\,\,\ka>-1/2,\\ \\
  \\
  \frac{1}{2}(f(x+y)+f(x-y)),\quad \text{if}\,\, \ka=-1/2.
  \end{array} \right.
\end{equation}
With the help of this translation operator, one defines the  convolution
 of $f\in L^1_\ka(0,\iy)$ and $g\in L^p_\ka(0,\iy)$ for
  $p\in [1,\,\iy)$ as the element  $f*_\ka g$ of $L^p_\ka(0,\,\iy)$ given by
\begin{equation}
(f*_\ka g)(x):=\int_0^\infty (\tau^x_\ka f)(y)\,g(y)\sigma_\ka(dy),\quad \ka\geq-1/2.\label{c1}
\end{equation}
The following properties are obvious
\begin{itemize}
  \item  $\mathscr{F}_\ka (\tau^x_\ka f)(t)=\mathscr{J}_\ka(xt)\mathscr{F}_\ka f(t),$
  \item  $\mathscr{F}_\ka(f*_\ka g)(x)=\mathscr{F}_\ka f(x) \mathscr{F}_\ka g(x).$
\end{itemize}
\section{Continuation of Radial Positive Definite Functions}
In this section, we will delve into the concept of radial positive definite functions, with a particular focus on the renowned Bochner and Schoenberg theorems. Additionally, we explore the continuation of radial definite positive functions.

A complex-valued function $f$ defined on $\mathbb{R}^n$ is said to be  definite positive and belongs to the class $\Phi(\mathbb{R}^n)$ if it is continuous at the origin and the matrix \cite{Berg,Sc,Rudin} $$\left(f(x_i-x_j)\right)_{i,j=1}^N$$ is nonnegative definite for all finite systems of points $x_1, x_2, \dots, x_N \in \mathbb{R}^n$ .\\
The classical Bochner's theorem \cite{Rudin} provides a fundamental characterization of the class $\Phi(\mathbb{R}^n)$. Specifically, a function $f$ belongs to  $\Phi(\mathbb{R}^n)$ if and only if it can be expressed as the Fourier transform of a finite non-negative Borel measure $\mu$ on $\mathbb{R}^n$ as:
\begin{equation}\label{Bochner}
f(x) = \int_{\mathbb{R}^n} e^{-i \langle x, t \rangle} \mu(dt),
\end{equation}
where $x,\,t\,\in \mathbb{R}^n$ and $\langle x, t \rangle$ denotes the inner product between $x$ and $t$.

In the context of radial functions, let us recall that a function $f$ defined on $\mathbb{R}^n$ is considered radial if there exists an even function $f_0$ defined on $\mathbb{R}$ such that $f(x) = f_0(\|x\|)$ for all $x \in \mathbb{R}^n,$ where $\|\cdot\|$ denotes the Euclidean norm on $\R^n$. In other words, the value of the radial function $f(x)$ depends solely on the magnitude or norm of the vector $x$.
By considering the properties of radial functions, we obtain the following expression for the Fourier transform:
\begin{align*}
\mathcal{F}f(\xi) &= \frac{1}{(2\pi)^n}\int_{\mathbb{R}^n} f(x)e^{-ix\cdot\xi}dx, \quad \xi \in \mathbb{R}^n \\
&= \frac{2\pi^{n/2}}{\Gamma(n/2)}\int_0^\infty f_0(r)\mathscr{J}_{n/2-1}(\|\xi\| r) r^{n-1}dr.
\end{align*}
A function $f: [0,\infty) \rightarrow \mathbb{R}$ is considered a radial positive definite function of the class $\Phi_n,$ if the function $f(\|\cdot\|)$ belongs to the class $\Phi(\mathbb{R}^n)$. The class $\Phi_n$ is characterized by Schoenberg theorem, which is referenced as \cite{Sc}.
\begin{theorem}\cite{Cholowinski} Function $f(\cdot)$ belongs to the class $\Phi_n$ if and only if
\begin{equation}\label{Bochner}
f(x)=\int_{0}^{\iy} \mathscr{J}_{n/2-1}(t x)\mu(dt),\end{equation}
where $\mu$ is a non-negative finite Borel measure on $[0,\,\iy)$.
\end{theorem}
In \cite{Cholowinski}, the authors introduced an innovative class of positive definite functions that relies on the generalized translation of Delsarte $\tau_\ka^{x}$, as  in equation \eqref{pB}.
\begin{definition}\cite{Cholowinski}. $\mathscr{P}_{\ka}$ is the set of continuous functions $f: [0;\infty) \rightarrow \mathbb{R}$ such that  the matrix $$\left(\tau^{x_i}_{\ka}f(x_j)\right)_{i,j=1}^N$$ is nonnegative definite for all finite systems of points $x_1, x_2, \dots, x_N \in [0,\,\iy)$.
\end{definition}Note that the discrete condition in the aforementioned definition implies the continuity of its integral counterpart function $f(x)$ on $[0, \infty)$. A bounded function $f(x)$ is said to be positive definite if for every $\varphi \in D_0(\mathbb{R})$ (where $D_0(\mathbb{R})$ denotes the space of even $C^{\infty}$ functions on $\mathbb{R}$ with compact support), the following inequality holds:

\begin{equation}\label{definite}
\int_{0}^{\infty}\int_{0}^{\infty} \tau^t_\ka f(x)\varphi(x)\varphi(t) \sigma_\kappa(x) \sigma_\kappa(t) \geq 0.
\end{equation}
For more details see \cite{Cholowinski,Chebli}.

In \cite{Cholowinski}, we found the following characterization of positive definite
functions for the Bessel operator, see also \cite{Trimeche}.

\begin{theorem}\cite{Cholowinski} A continuous function $f(x)$ is a bounded positive definite
function for the Bessel operator $\mathscr{B}_\nu$ if and only if there exists  a non-negative finite Borel measure $\mu$ on $[0,\,\iy)$ such  that for every $x\geq 0$
\begin{equation}\label{Bochner}
f(x)=\int_{0}^{\iy} \mathscr{J}_\ka(tx)\mu(dt).
\end{equation}
\end{theorem}
In the following proposition we will show that when $\ka$ is half-integer ($\ka=n/2-1$) the class $\mathscr{P}_\kappa$ coincides with $\Phi_n$. \begin{proposition}For every $n\in \mathbb{N},$ we have $\mathscr{P}_{n/2-1}=\Phi_n.$
\end{proposition}
\begin{proof}Let $f$ be a integrable radial function $f(x)=f_0(\|x\|)$ on $\R^n$ . Since the Lebesgue measure is invariant under the orthogonal transformation, then the function
\begin{equation*}
x\rightarrow\int_{\R^n}f(x-y)dy
\end{equation*}
is a radial function. Therefore
\[
\int_{\R^n}f(x-y)dy=\int_{\R^n}f(\|x\|e_1-y)dy,
\]
where $e_1=(1,\,0,\,\dots,\,0).$
It follows
\begin{align*}
\int_{\R^n}f(x-y)dy&
=\int_{\R}\int_{\R^{n-1}}f_0(\sqrt{(\|x\|+p)^2+\|y\|^2})dy dp\\&
=\omega_{n-2}\int_{\R}\int_{0}^{\iy}f_0(\sqrt{(\|x\|+p)^2+ \rho^2})\rho^{n-2}d\rho \,dp.
\end{align*}
where $$\omega_{n-1}=\frac{2\pi^{n/2}}{\Gamma(n/2)}.
$$
Making the substitution $p=t\cos \theta,$ $\rho=tsin\theta$
\begin{align*}
\int_{\R^n}f(x-y)dy=&
\omega_{n-2}\int_{0}^{\iy}\int_{0}^{\pi}f_0(\sqrt{\|x\|^2+t^2+ 2t\|x\|\cos\theta} )\sin ^{n-2}(\theta)\,t^{n-1}\,dt\\&=
\int_{0}^{\iy}\tau^{\|x\|}_{n/2-1}f_0(r) \sigma_{n/2-1}(dr).
\end{align*}
The result follows  from \eqref{definite} and Proposition 6.4 in \cite{We}.
\end{proof}

\section{Characterization of positive definite functions via Caputo fractional derivatives }
In this section, we present a characterization of the representation of functions on the interval $(0, \infty)$ as Fourier-Bessel transforms using Riemann-Liouville integral and Caputo fractional derivative.\\
To begin, let us recall the definitions of left-sided and right-sided fractional Riemann-Liouville integrals of order $\alpha$ \cite{Kilbas}.

For $\mathrm{Re}(\alpha) > 0$, the left-sided fractional Riemann-Liouville integral of order $\alpha$ is given by:
\begin{equation}\label{Rie1}
I_+^\alpha \{f(t);x\}: = \frac{1}{\Gamma(\alpha)}\int_{0}^{x}\frac{f(t)dt}{(x-t)^{1-\alpha}} \quad (x > 0).
\end{equation}
Similarly, the right-sided fractional Riemann-Liouville integral of order $\alpha$ is defined as:
\begin{equation}\label{eq:right_integral}
I_-^\alpha\{ f(t);x\} := \frac{1}{\Gamma(\alpha)}\int_{x}^{\infty}\frac{f(t)dt}{(t-x)^{1-\alpha}} \quad (x \geq 0).
\end{equation}
These integrals involve the gamma function $\Gamma(\alpha)$, which is defined as the integral:
\begin{equation}
\Gamma(\alpha) = \int_0^\infty t^{\alpha-1}e^{-t}dt \quad \alpha > 0.
\end{equation}
When $\alpha=n$ is a positive integer we have
\begin{align*}
I^n_+\{f(t);x\}&=\frac{1}{(n-1)!}\int_{0}^{x}f(t)(x-t)^{n-1}\,dt\\&
=\int_{0}^{x}dx_1\int_{0}^{x_1}dx_2\dots \int_{0}^{x_{n-1}}f(x_n)\,dx_{n}.
\end{align*}
The following semigroup property holds
\begin{equation}\label{ER1}
I^{\al}_{+}I^{\beta}_{+}=I^{\al+\beta}_{+}, \quad \Re(\al),\,\Re(\beta)>0.
\end{equation}
For $\Re(\alpha)\geq 0$ and $\alpha \notin \mathbb{N}$, the Caputo fractional derivative $D^\alpha_* f$ is defined as \cite[\S 2]{Kilbas}:
\begin{equation}
\mathcal{D}^\alpha_* \{f(t);x\} := I^{n-\alpha}_+ \{D^n f(t);x\}, \quad n=[\Re(\alpha)]+1,
\end{equation}
where $I^{n-\alpha}_+$ represents the Riemann-Liouville fractional integral defined in \eqref{Rie1} and $D^n$ denotes the $n$-th derivative with respect to $x$.
For $\alpha\in \mathbb{N}$, the Caputo fractional derivative is given by $\mathcal{D}^n  = D^n$.\\
A more general fractional operators are  the left-side and right-sided Erdelyi-Kober
integrals that involve an additional parameter, known as the Erd\'{e}lyi-Kober parameter. Let $f$ be continuous function on $[0,\iy), $  $\mathrm{Re}(\alpha) > 0,$ and $\eta > -\frac{1}{2}$. The left-sided Erd\'{e}lyi-Kober fractional integral of order $\alpha$ and parameter $\eta$, denoted by $I_{+,2,\eta}^\alpha f(x)$, is given by:
\begin{align}\label{ko1}
&(I^\alpha_{+,2,\eta}f)(x):=\frac{2}{\Gamma(\al)}\,\frac{1}{ x^{2( \alpha+\eta)}}\,\int_{0}^{x}f(t)(x^2-t^2)^{\alpha-1}\,t^{2\eta+1}\,dt\quad x>0.
\end{align}
Similarly, the right-sided Erd\'{e}lyi-Kober fractional integral of order $\alpha$ and parameter $\eta$, denoted by $I_{-,2,\eta}^\alpha f(x)$, is defined as:
\begin{align}\label{ko2}
(I^\alpha_{-,2,\eta}f)(x):=\frac{2 }{\Gamma(\al)}\,x^{2\eta}\,\int_{x}^{\iy}f(t)(t^2-x^2)^{\alpha-1}t^{1-2\al-2\eta}dt\quad (x\geq0).
\end{align}
To handle even continuous functions $f$ defined on the real line $\mathbb{R}$, we can extend the definition of $I^\alpha_{\pm,2,\eta}f$ by utilizing parity. Specifically, we define $I^\alpha_{\pm,2,\eta}f$ on $\mathbb{R}$ by setting $$(I^\alpha_{\pm,2,\eta}f)(x) = (I^\alpha_{\pm,2,\eta}f)(|x|)\quad \text{for all}\,\, x\in \R.$$
A straightforward computation shows that
\begin{align}\label{Erd1}
 &I^{\al+1/2}_{+,-1/2} f(x)=x^{-2\al}I^{\al+1/2}_+\big\{
 (\tfrac{f(\sqrt{t})}{\sqrt{t}},x^2\big\}, \\&I^{\al+1/2}_+ f(x)=x^{\al}I^{\al+1/2}_{+,-1/2}l
 \big\{t\,f(t^2),\sqrt{x}\big\}.
\end{align}
The following theorem represents the primary result in this section
\begin{theorem} Let $\ka\geq -1/2$ and $\epsilon>0,$ the following hold
\begin{itemize}
\item [(i)]  $\mathscr{P}_{\ka+\epsilon}\subset \mathscr{P}_\ka.$
\item [(ii)] $f\in \mathscr{P}_\ka$ if and only if $I^\epsilon_{+,2,\ka}f\,\in \mathscr{P}_{\ka+\epsilon}.$
\end{itemize}
Furthermore, the mapping
$I^\epsilon_{+,2,\ka}:\,\mathscr{P}_\ka \to \mathscr{P}_{\ka+\varepsilon}$ is one-to-one.
\end{theorem}
\begin{proof}By utilizing the Sonine integral representation for the Bessel function, as described in \cite[\S 12.11]{8}, we obtain the following result
$$I^\epsilon_{+,2,\ka}\{\mathscr{J}_\ka(u);t\}=\frac{\Gamma(\ka+1)}{\Gamma(\ka+\epsilon+1)}
\mathscr{J}_{\ka+\epsilon}(t).$$
Now, let $f\in \mathscr{P}_{\ka+\epsilon},$ then there  a non-negative finite Borel measure $\mu$ on $[0,\,\iy)$ such that
\begin{equation*}
f(x)=\int_{0}^{\iy}\mathscr{J}_{\ka+\epsilon}(xt)d\mu(t).
\end{equation*}
It follows
\begin{align*}
f(x)&=\frac{\Gamma(\ka+\epsilon+1)}{\Gamma(\ka+1)}\int_{0}^{\infty}I^\epsilon_{+,2,\ka}\{\mathscr{J}_\ka(u);t\}d\eta(t)\\&
\\&=\int_{0}^{\iy} \mathscr{J}_\ka(xu) d\widetilde{\eta}(u),
\end{align*}
where $$d\widetilde{\eta}(u)=\frac{2\Gamma(\ka+1)}{\Gamma(\ka+1)\Gamma(\epsilon)}\,u^{2(\ka+1)}\,
\int_{u}^{\iy}t^{-2{\ka+\epsilon}}{(t^2-u^2)^{\epsilon-1}}
\,d\eta(t).$$
Therefore $f\in \mathscr{P}_\ka.$ This proves $(i)$ and $(ii).$
To complete the proof, we aim to demonstrate that if $f$ is a bounded measurable function on $[0, \infty)$ and $\alpha, \beta > -1$, such that the integral $\int_{0}^{1}t^\alpha(1-t^2)^\beta f(xt) dt = 0$, then we can conclude that $f = 0$ almost everywhere.
By using the transformation $t \rightarrow \sqrt{t}$, we can rewrite the integral as follows:
$$\int_{0}^{1}t^\alpha(1-t^2)^\beta f(xt) dt =\frac{1}{2}\int_{0}^{1}t^{(\alpha-1)/2}(1-t)^\beta f(x\sqrt{t}) dt .$$
Applying the Titchmarsh theorem on convolution (see, \cite{Trigueb}) to this transformed integral, we can establish the desired result.
\end{proof}
The following theorem, which serves as the second main result in this section, is inspired by the work of R. M. Trigub \cite[\S 6.3]{Trigueb}.
\begin{theorem}
Let $\ka>-\frac{1}{2}.$ In order to $f\in \mathscr{P}_{\ka},$ is necessary and sufficient  that the function $f(\cdot)$  satisfies:
\begin{itemize}
\item [(i)] $\big(x^{\ka}f(\sqrt{x})\big)^{(j)}=0,$ for $j=0,\,2,\,\dots,\,n,$
\item [(ii)]$ xD\,^CD_0^{\ka-1/2}\{t^\ka f(\sqrt{t}),x^2 \}\in \mathscr{P}_{-1/2},$
\end{itemize}
where $\ka=n+r$ with $-1/2 < r \leq 1/2.$
\end{theorem}
\begin{proof}
Necessity:
Suppose $f\in\mathscr{P}_{\kappa}$. By Theorem 4.10, $f\in\mathscr{P}_{\kappa}$ if and only if there exists $g\in\mathscr{P}_{-1/2}$ such that
\begin{equation}\label{Es1}
f(x)=I^{\kappa+1/2}_{+,2,-1/2}\{g(t);x\} =\frac{2}{\Gamma(\kappa+1/2)x^{2\kappa}}\int_{0}^{x}g(t)(x^2-t^2)^{\kappa-1/2}dt.
\end{equation}
Using \eqref{Erd1}, we can rewrite the equation \eqref{Es1} in the following equivalent form:
\begin{equation*}
x^{\kappa}f(\sqrt{x}) = \frac{1}{\Gamma(\kappa + 1/2)}\int_{0}^{x}g(\sqrt{t})(x-t)^{\ka-1/2}t^{-1/2}dt,
\end{equation*}
Since the function $t\to\frac{g_0(t)}{\sqrt{t}}$ is locally integrable on $[0,\infty)$, it follows that $x\to x^{\kappa}f(\sqrt{x})$ belongs to the class $C^n$ on $[0,\infty),$ and  for $j=0,1,\dots,n$, we have
\begin{equation*}
\left[\frac{d^j}{dx^j}\left(x^{\kappa}f(\sqrt{x})\right)\right]_{x=0}=0,
\end{equation*}
where $$\kappa = n+r, \quad \text{and}\quad -\frac{1}{2}<r\leq\frac{1}{2}.$$
Furthermore
\begin{itemize}
  \item for $r=1/2,$ we have
  \begin{equation*}
  g(\sqrt{x})=\sqrt{x}\Big[x^{n+1/2}f(\sqrt{x})\Big]^{(n+1)}=\mathcal{D}^{n+1}\Big[x^{n+1/2}f(\sqrt{x})\Big],
  \end{equation*}
  \item for  $-1/2 <r <1/2$
  \begin{equation}\label{2}
  \frac{d^n}{dx^n}\Big[x^\ka f(\sqrt{x})\Big]=\frac{1}{\Gamma(r+1/2)}\int_{0}^{x}g(\sqrt{t})(x-t)^{r-1/2}t^{-1/2}\,dt.
\end{equation}
\end{itemize}
In this case we can solve the above Abel integral equation to get
\begin{align*}
g(\sqrt{x})&=\frac{\sqrt{x}}{\Gamma(\frac{1}{2}-r)}\frac{d}{dx}\int_{0}^{x}\frac{d^n}{dt^n}\Big(t^\ka
f(\sqrt{t})\Big)\frac{dt}{(x-t)^{r+1/2}},\\&=\sqrt{x}D\,^CD_0^{\ka-1/2}\{t^\ka f(\sqrt{t});x\}.
\end{align*}
This proves $(i)$ and $(ii).$

Sufficiency: Let us assume that $f$ satisfies conditions $(i)$ and $(ii)$. We consider two cases based on the value of $\kappa = n+r$, where $-1/2 < r \leq 1/2$.

Case 1: $r=1/2$
From condition $(ii)$, we have
\begin{equation}
g(\sqrt{x}) = \sqrt{x}\left[\left(x^{n+1/2}f(\sqrt{x})\right)^{(n+1)}\right].
\end{equation}
Applying $I^{n+1}_+$ to the above equation and using condition $(i)$, we obtain
\begin{equation}\label{eq1}
f(x) = \frac{2}{n!}\cdot\frac{1}{x^{2n+1}}\int_{0}^{x}(x^2-t^2)^n g(t)dt \in \mathcal{P}_{n+1/2}.
\end{equation}

Case 2: $-1/2 < r < 1/2$  From condition $(ii)$, we can write
\begin{equation*}
  \frac{d^n}{dx^n}\Big[x^\ka f(\sqrt{x})\Big]=\frac{1}{\Gamma(r+1/2)}\int_{0}^{x}g(\sqrt{t})(x-t)^{r-1/2}t^{-1/2}\,dt.
\end{equation*}
Applying $I_+^n$ to the above equation and using condition $(i)$ and semigroup property for Riemann-Liouville integral \eqref{ER1}, we obtain
\begin{equation}\label{eq2}
f(x) = I^{\kappa+1/2}_{+,2,-1/2}\{g(t);x\} \in \mathcal{P}_{\kappa}.
\end{equation}

Therefore, in both cases, we have shown that $f \in \mathcal{P}_{\kappa}$, which completes the proof of sufficiency.
\end{proof}

\section{Characterization via complete monotone function}
In this section, we explore the characterization of the class $\mathcal{P}_\ka$ by leveraging Bernstein's theorem for completely monotone functions. \\
Recall that, a function $f:\,[0,\,\infty)\rightarrow \mathbb{R}$ is  completely monotone function, if it is of class $C^\iy$ on $(0,\,\iy)$ and
 $$(-1)^{n}f^{n}(x)\geq0,\quad x>0,\,n\in \mathbb{N}.$$
 By Bernstein characterization, $f$ is  completely monotone functions, if and only if,  there exists some measure $\mu$ on  $[0,\,\iy)$ such that
 $$f(x)=\int_{0}^{\iy}e^{-tx}d\mu(t).$$
 We now focus on the connection between positive definite radial functions and completely monotone functions, as originally established by Schoenberg in 1938, see \cite{Sc}.
\begin{theorem} \cite{Cholowinski} A function $f$ is completely monotone on $[0,\infty)$ if and only if the function $f(x) := f(\|x\|)$ is positive definite on every $\mathbb{R}^n$.
\end{theorem}
The main result in this section is the following theorem
\begin{theorem}
For any $\ka \geq -1/2$, a function $f$ belongs to the class $\mathscr{P}_{\ka}$ if and only if $f(\sqrt{x})$ is completely monotone on the interval $[0, \infty)$.
\end{theorem}
\begin{proof}Sufficiency: By the Bernstein theorem, there exists a finite positive measure $\mu$ on $[0, \infty)$ such that for all $x \geq 0$,
\begin{equation*}
f(x) = \int_{0}^{\infty} e^{-tx^2} d\mu(t).
\end{equation*}
On the other hand, according to \cite[formula 4.11.27]{An}, we have
\begin{equation}
e^{-tx^2} = \int_{0}^\infty \mathscr{J}_\kappa(ux) \frac{e^{-\frac{u^2}{4t}}}{(2t)^{\kappa+1}} \sigma\kappa(du).
\end{equation}
Using this representation, we can rewrite $f(x)$ as
\begin{equation*}
f(x) = \int_{0}^{\infty} \mathscr{J}_\kappa(tx) d\widetilde{\mu}(t),
\end{equation*}
where $$d\widetilde{\mu}(t) = \int_{0}^\infty \frac{e^{-\frac{x^2}{4t}}}{(2t)^{\kappa+1}} d\mu(t) \sigma_\kappa(du).$$
Therefore, we have shown that $f(x) \in \mathcal{P}_\kappa$ for all $\kappa \geq -1/2$. Hence, the sufficiency of the conditions is established.\\
Necessity: Suppose $f \in \mathcal{P}_\kappa$ for all $\kappa \geq -1/2$. In particular, $f \in \Phi_n$ for all $n \in \mathbb{N}$. Then the result follows from Theorem 5.1, which establishes the connection between positive definite radial functions and completely monotone functions.
Hence, the necessity of the conditions is demonstrated.
\end{proof}

\section{Application: Positivity of the Fundamental Solution}
In this section, we will show the positivity of the fundamental solution of the following space-fractional Bessel diffusion equation
\begin{equation}
\left\{
  \begin{array}{l l}\label{e2}
       \,\partial_t
 u(t,x)=-(-  \mathscr{B}_\ka)^{\alpha/2}u(t,x),\\
      u(0,x)=f(x),\,\, x\geq 0,\quad t>0.\end{array} \right.
\end{equation}
Here, $(-\mathscr{B}_\kappa)^{\alpha/2}$ is the fractional Bessel operator, which is given by \cite{Bou}
\begin{equation*}
(-\mathscr{B}_\kappa)^{\alpha/2}f(x)=\frac{2^{\alpha+\ka}\Gamma(\ka+\frac{\alpha}{2}+1)}
{\Gamma(\ka+1)
|\Gamma(-\frac{\alpha}{2})|}\int_{0}^{\infty}\frac{f(x)-\tau^x_\ka f(y)}{y^{\alpha+1}}\,dy.
\end{equation*}
Let's denote the Fourier-Bessel transform of a function $u(t,x)$ with respect to $x$ as $\hat{u}(t,\lambda)$, where $\lambda\geq 0$. Applying the Fourier-Bessel transform to both sides of the equation in \eqref{e2}, we obtain:
\begin{equation*}
\left\{
  \begin{array}{l l}\label{d2}
       \,\partial_t  \widehat{u}(t,\lambda)=- \lambda^\alpha \widehat{u}(t,\lambda),\\
      \widehat{u}(0,x)=\widehat{f}(\lambda).\end{array} \right.
\end{equation*}
Then \[\widehat{u}(\lambda,t)= \widehat{f}(\lambda)e^{-\lambda^\alpha t}.\]
Therefore
\begin{equation*}
u(t,x)=(\mathscr{G}^{\alpha,\ka}_t*f)(x),
\end{equation*}
where
\begin{equation}\label{gf1}
\mathscr{G}^{\alpha,\ka}_t(x)=\mathscr{G}^{\alpha,\ka}(x,t)=\int_{0}^{\infty}e^{-\lambda^{\alpha}
 t }\mathscr{J}_{\ka}(\lambda x)\,\sigma_\ka(d\lambda).
 \end{equation}
 Using the following scaling rules for the Fourier-Bessel transform:
\begin{align*}
\int_{0}^{\infty} f(a x)\mathscr{J}_\ka(\lambda x)\sigma_\ka(dx)=  \frac1{a^{2\ka+2}}\int_{0}^{\infty} f(x)\mathscr{J}_\ka(\lambda x/a)\sigma_\ka(dx),\quad a>0,
\end{align*}
we  obtain the following scaling property of the kernel $\mathscr{G}^{\alpha,\ka}(x,t)$
    \begin{equation*}
      \mathscr{G}^{\alpha,\ka}(t,x)=t^{-2(\ka+1)/\alpha}\mathscr{G}^{\alpha,\ka}(xt^{-1/\alpha},1), \quad t>0,\quad x\in \mathbb{R}.
    \end{equation*}Consequently, introducing the similarity variable $x/t^\alpha,$ we can write
    \begin{equation*}
      \mathscr{G}^{\alpha,\ka}(x,t)=t^{-2(\ka+1)/\alpha}\mathscr{K}^{\alpha,\ka}(xt^{-1/\alpha}),
    \end{equation*}
where \begin{equation}\label{B22}
\mathscr{K}^{\alpha,\ka}(x)=\int_{0}^{\iy}e^{-\lambda^\alpha}\mathscr{J}_\ka(\lambda x)\sigma_\ka(d\lambda).
\end{equation}

 $\mathrm{Particular\,\,\, cases}$ of the density $\mathscr{K}^{\alpha,\ka}$ are the following:
\begin{itemize}
  \item The  density $\mathscr{K}^{2,\ka}(x), \, \ka \geq -1/2$, is the Gaussian density kernel
\begin{equation}
\mathscr{K}^{2,\ka}(x)=\frac{e^{-\frac{x^2}{4}}}{2^{\ka+1}}.
\end{equation}
\item The  density $\mathscr{K}^{1,\ka}, \; \ka \geq -1/2$, is  the Poison density
\begin{equation}
\mathscr{K}^{1,\ka}(x)=\frac{2^{\ka+1}\Ga(\ka+\tfrac{3}{2})}{\sqrt{\pi}}\frac{1}
{(1+x^2)^{\ka+\frac{3}{2}}}.
\end{equation}
\end{itemize}
\begin{proposition}\cite{Bou} The following holds.
\begin{enumerate}
\item For $\ka \geq -1/2$,  $0<\alpha< 2$ , we have
\begin{itemize}
  \item  $x\neq0$
   \begin{equation*}
\mathscr{K}^{\alpha,\ka}(x)=\frac{1}{2^{\ka+1}\pi}\sum_{n=1}^{\infty}
(-1)^{n+1}\frac{\Ga(1+\alpha n/2)\Ga(\ka+1+\alpha n/2)}{n!}\sin(\frac{\alpha n\pi}{2})\,
\left(\frac{4 }{x^2}\right)^{\frac{\alpha n}{2}+\ka+1}.
\end{equation*}
\item $x=0$, $\mathscr{K}^{\alpha,\ka}(0)=\frac{\Ga(\frac{2(\ka+1)}{\alpha})}{\alpha}$.
\end{itemize}
\item For $\ka \geq -1/2$ and $\alpha>1,$ we have
 \begin{equation}\label{series}
\mathscr{K}^{\alpha,\ka}(x)=\frac{1}{\alpha 2^{\ka}}
\sum_{n=0}^{\infty}\frac{(-1)^n}{n!}\frac{\Ga(\tfrac{2}{\alpha}(n+\ka+1))}{\Ga(\ka+1+n)} \left(\frac{x^2}{4}\right)^{n}.
\end{equation}
\end{enumerate}
\end{proposition}
\begin{lemma}
Let $v$ be an even function of class $C^\infty$ with compact support in $[-1,1]$ that is positive and satisfies $\|v\|_{1,\kappa}=1$. For any $\varepsilon\in (0,1]$, define the function $v_\varepsilon(x)=\varepsilon^{-2\kappa-2}v\left(\frac{x}{\varepsilon}\right)$. Then $v_\varepsilon$ has compact support in $[-\varepsilon,\varepsilon]$ and $\|v_\varepsilon\|_{1,\kappa}=1$.
 Furthermore, $$|\widehat{v_\varepsilon}(\lambda)-1|\leq \varepsilon \lambda\quad \text{ for all}\,\, \lambda \geq 0.$$
\end{lemma}
\begin{proof}
First, we note that $v_\varepsilon$ has compact support in $[-\varepsilon,\varepsilon]$ since $v$ has compact support in $[-1,1]$. To show $\|v_\varepsilon\|_{1,\kappa}=1$, we compute
\begin{align*}
\|v_\varepsilon\|_{1,\kappa} = \int_{0}^\varepsilon v_\varepsilon(x)\sigma_\ka(dx)
&= \int_{0}^1 v(x) \sigma_\ka(dx)=\|v\|_{1,\ka}=1.
\end{align*}
It follows
\begin{align*}
\widehat{v_\varepsilon}(\lambda)-1 &= \int_{0}^\varepsilon v_\varepsilon(x)(\mathscr{J}_\ka(\lambda x)-1) \sigma_\ka(dx)
\end{align*}
Since $$|\mathscr{J}_\ka(x)-1|\leq x \quad \text{for all}\,\, x\geq 0,$$ we have
\begin{align*}
|\widehat{v_\varepsilon}(\lambda)-1| \leq \varepsilon\lambda \int_0^\varepsilon v_\varepsilon(x)\sigma_\ka(dx)=\varepsilon\lambda.
\end{align*}
Therefore, the lemma holds.
\end{proof}
\begin{proposition}
A continuous bounded function on $[0,\infty)$ is positive definite if and only if for every even function $h$ in the Schwartz space $S(\mathbb{R})$, the inequality $\widehat{h}(x)\geq 0 \quad \text{for all}\,\,x\geq 0$ implies $\widehat{hf}(x) \geq 0\quad \text{for all}\,\,x\geq 0$.
\end{proposition}
For the proof of this proposition, refer to the last remark in \cite{Cholowinski}.\\

The following theorem represents the main result of this section:
\begin{theorem}
For $\ka\geq -1/2$ and $0 < \alpha\leq 2$, we have $\mathscr{G}^{\alpha,\ka}(x,t)\geq 0$.
\end{theorem}
\begin{proof}
Since the function $e^{-\lambda^\alpha}$ is completely monotone for $0<\alpha<2$, it follows from Theorem 5.2 that it is positive definite for all $\kappa \geq -1/2$. Using the notation of Lemma 6.10 we have
\[v_\varepsilon(x)=\int_{0}^{\iy}\widehat{v_\varepsilon}(\lambda)\sigma_\ka(d\lambda).\]
Since $\widehat{v_\varepsilon}(\lambda) \in S(\mathbb{R})$ is even and positive, and the function $e^{-\lambda^\alpha}$ is positive definite according to the above proposition, we have
\begin{align*}\int_{0}^{\iy}\widehat{v_\varepsilon}(\lambda)e^{-\lambda^\alpha } \mathscr{J}_\ka(\lambda x)\sigma_\ka(d\lambda)\geq 0.
\end{align*}
The lemma 6.8 shows that this integral converges to $\mathscr{K}^{\alpha,\kappa}_t(x),$ as $\varepsilon\rightarrow 0$. This completes the proof of the theorem.
\end{proof}
\section{ Concluding Remarks}
In conclusion, this study has made significant contributions to the field by investigating the continuation and characterization of radial positive definite functions beyond integer dimensions. By utilizing fractional derivatives and exploring their relationship with non-integer continuation, a deeper understanding of these functions has been achieved.

The practical relevance of the findings has been demonstrated through the investigation of the space fractional diffusion equation. By applying the obtained characterizations, the positivity of the fundamental solution has been established. This showcases the practical implications and real-world applications of continued radial positive definite functions.

Overall, this research expands the knowledge and understanding of radial positive definite functions, paving the way for further advancements in the field. The insights gained from this study have the potential to impact various mathematical disciplines, as well as practical domains such as image processing, data analysis, and machine learning. By extending the study beyond integer dimensions, new avenues for research and applications are opened, contributing to the advancement of the field as a whole.
\\

$\bf{Acknowledgment.}$ The authors extend their appreciation to the Deputyship for Research and Innovation, Ministry of Education in Saudi Arabia for funding this research work through the project no ( IFKSUOR3-368-1).



\end{document}
=\frac{\Gamma(-\gamma)\Gamma(k+1/2)}{\Gamma(1/2)\Gamma(k)}
\int_{-1}^{1}(it\lambda)^{\gamma}(1-t)^{k-1}(1+t)^k\,dt
\end{align}
we have
\begin{align}
\int_{0}^{\iy}\frac{1-\mathcal{E}_k(-i\lambda y)}{y^{1+\gamma}}\,dy
&
=\frac{\Gamma(-\gamma)\Gamma(k+1/2)}{\Gamma(1/2)\Gamma(k)}
\Big\{(i\lambda)^\gamma\int_{0}^{1}t^{\gamma}(1-t)^{k-1}(1+t)^k\,dt\\&+
(-i\lambda)^\gamma\int_{0}^{1}t^{\gamma}(1+t)^{k-1}(1-t)^k\,dt\Big\}\\&=
\frac{\Gamma(-\gamma)\Gamma(k+1/2)}{\sqrt{\pi}}|\lambda|^\gamma\Big\{
\frac{\Gamma(\frac{\gamma+1}{2})}{\Gamma(k+\frac{\gamma+1}{2})}
\cos(\frac{\gamma\pi}{2})+i\,sgn(\lambda)\,\frac{\Gamma(\frac{\gamma+2}{2})}
{\Gamma(k+\frac{\gamma+2}{2})}
\sin(\frac{\gamma\pi}{2})\Big\}
\end{align}
Using Beta function
\begin{equation}\label{b}
  \frac{\Gamma(x)\Gamma(y)}{\Gamma(x+y)}=\int_{0}^{1}t^{x-1}(1-t)^{y-1}\,dt
\end{equation}
we obtain
\begin{equation}
  \int_{0}^{1}t^{\gamma}(1-t)^{k-1}(1+t)^k\,dt=\frac{1}{2}\frac{\Gamma()}{\Gamma()}
\end{equation}
